Author's response to reviews

Title: Cost of Stroke: A Controlled National Study Evaluating Societal Effects on Patients and Their Partners

Authors:

Poul J Jennum (poul.joergen.jennum@regionh.dk)
Helle K Iversen (helle.klinigenberg.iversen@regionh.dk)
Rikke Ibsen (rikke@i2minds.dk)
Jakob Kjellberg (Jakj@kora.dk)

Version: 5 Date: 29 July 2015

Author's response to reviews:

Editors Comment:

- Please move Authors Contribution, Competing Interest, Acknowledgment and Tables to the correct places.
- Delete funding section and add funding info to Acknowledgment.
- Do not upload cover letters and copies of the manuscript as additional files.

The above points are corrected.